Author’s response to reviews

Title: Estimate of undergraduate university student alcohol use in China: a systematic review and meta-analysis

Authors:

Ian Newman (inewman1@unl.edu)
Lanyan Ding (dinglanyan@gmail.com)
Yonghua Feng (zhxfyh@hotmail.com)

Version: 1 Date: 07 Jul 2017

Author’s response to reviews:

The authors are grateful to the editor and the reviewers for their detailed comments. We have incorporated all the changes requested. The 3-page point-by-point response can be found at the beginning of the revised manuscript, before the title page. Again thank you for the insightful and detailed comments and suggestions.